Author's response to reviews

Title: The REFLO-STEMI Trial comparing intracoronary Adenosine, Sodium Nitroprusside and standard therapy for the attenuation of infarct size and microvascular obstruction during primary percutaneous coronary intervention: study protocol for a randomized controlled trial.

Authors:

Sheraz A Nazir (sn191@le.ac.uk)
Jamal N Khan (jk211@le.ac.uk)
Islam Z Mahmoud (islam.mahmoud@kcl.ac.uk)
John P Greenwood (j.greenwood@leeds.ac.uk)
Daniel J Blackman (daniel.blackman@leedsth.nhs.uk)
Vijay Kunadian (vijay.kunadian@newcastle.ac.uk)
Martin Been (vijay.kunadian@newcastle.ac.uk)
Keith R Abrams (kra1@le.ac.uk)
Robert Wilcox (robert.wilcox@nottingham.ac.uk)
Jennifer Adgey (jennifer.adgey@belfasttrust.hscni.net)
Gerry P McCann (gerry.mccann@uhl-tr.nhs.uk)
Anthony H Gershlick (agershlick@aol.com)

Version: 5 Date: 3 September 2014

Author's response to reviews:

Dear Editors

I have added a middle initial for author Daniel Blackman as per his request.

Otherwise, the manuscript is unchanged.

Kind regards

Dr Sheraz Nazir